Author's response to reviews

Title: Intake of Fucus vesiculosus, an Edible Brown Seaweed, Affects the Menstrual Cycle and Hormonal Status of Pre-menopausal Women

Authors:

Dr Christine F. Skibola Ph.D. (chrisfs@uclink.berkeley.edu)

Version: 2 Date: 16 Jan 2004

Figures 1a and 1b have been replaced by Figures 1 and 2. Descriptions of the tables have been revised due to an error while cutting and pasting. The Results section referring to these figures has also been revised.